Author’s response to reviews

Title: Human Exposure to Nitro Musks and the Evaluation of their Potential Toxicity: An Overview

Authors:

Kathryn M Taylor (kmtaylor@hsph.harvard.edu)
Marc G Weisskopf (mweissko@hsph.harvard.edu)
James P Shine (jshine@hsph.harvard.edu)

Version: 3 Date: 30 September 2013

Author’s response to reviews: see over
Dear Editor,

On behalf of my coauthors I would like to submit our newly revised manuscript. We have made the changes that the editors have requested. Specifically we have:

- Inserted the heading ‘email addresses’ above the list of emails
- Corrected all headings so that only the first letter is capitalized
- Changed italicized text except for where indicated in the references
- Removed issue numbers from the references
- Corrected the headings for both: List of abbreviations and Authors' contributions
- Inserted the statement of no competing risks

We again thank you for considering this manuscript.

Kathryn Taylor, SM
Departments of Environmental Health
Harvard School of Public Health